Author's response to reviews

Title: Contralateral extradural hematoma following decompressive craniectomy for acute subdural hematoma: the value of ICP monitoring. Case presentation

Authors:

Lucas C Meguins (lucascrociati@hotmail.com)
Gustavo B Sampaio (gustavo@hotmail.com)
Eduardo C Abib (eduardo@hotmail.com)
Rodrigo R Adry (rodrigo@hotmail.com)
Richam F Ellakkis (richam@hotmail.com)
Filipe W Ribeiro (filipe@hotmail.com)
Angelo L Masset (angelo@hotmail.com)
Dionei F Morais (dionei.fm@terra.com.br)

Version: 5 Date: 5 February 2014

Author's response to reviews: see over
Response to the reviewers` comments:

Reviewer 1:
- The origin of the frontal bleeding was described on the manuscript
- No coagulopathy was found on admission (All laboratory exams were normal)

Reviewer 2:
- Concept: we included on the title "the importance of ICP monitoring"
- Contents: No coagulopathy was found on admission (All laboratory exams were normal). We improved the explanation on the pathophysiology involved in the formation of delayed contralateral EDH following decompressive craniectomy.
- Language / style: we made all the changes suggested by the reviewer regarding language corrections